Author's response to reviews

Title: Movements disorders in neuroleptic-naive patients with schizophrenia spectrum disorders in a low income setting

Authors:

Moges Ayehu (ayehumoges@yahoo.com)
Teshome Shibre (teshome.kelkile@gmail.com)
Barkot Milkias (baromilki@gmail.com)
Abebaw Fekadu (abe.wassie@kcl.ac.uk)

Version: 2 Date: 25 November 2013

Author's response to reviews:

Dear Sir, Dear Madam,

We are grateful to the editorial office for recognising the omissions promptly. In this revision the suggested revisions have been made. We have now include separate statements on competing interests, authors’ contributions and acknowledgement.

Your sincerely,

Dr Abebaw Fekadu, on behalf of the authors